

Author's response to reviews

Title: The Effect of Dams and Seasons on Malaria Incidence and Anopheles Abundance in Ethiopia

Authors:

Delenasaw Yewhalaw (delenasaw.yewhalaw@ju.edu.et)

Yehenew Getachew (yehenew.getachew@ju.edu.et)

Kora Tushune (kora.tushune@ju.edu.et)

Kifle W/Michael (bethy_kifle@yahoo.com)

Wondwossen Kassahun (kasahun.eba@ju.edu.et)

Luc Duchateau (Luc.Duchateau@Ugent.be)

Niko Speybroeck (Niko.Speybroeck@uclouvain.be)

Version: 2 **Date:** 26 December 2012

Author's response to reviews: see over